**Author's response to reviews**

**Title:** Ultra-early tranexamic acid after subarachnoid hemorrhage (ULTRA): study protocol for a randomized controlled trial

**Authors:**

Menno R Germans (M.R.Germans@amc.uva.nl)  
René Post (R.Post@amc.uva.nl)  
Bert A Coert (B.A.Coert@amc.uva.nl)  
Gabriël JE Rinkel (G.J.E.Rinkel@umcutrecht.nl)  
W Peter Vandertop (W.P.Vandertop@amc.uva.nl)  
Dagmar Verbaan (D.Verbaan@amc.uva.nl)

**Version:** 2  **Date:** 27 April 2013

**Author's response to reviews:**

Dear Editor-in-Chief,

I am delighted to hear that the protocol of our study is accepted for publication in Trials. Below is an overview of the changes that have been made.

- The article title has been changed into the desired format
- The study protocol is formatted as outlined in the instructions for authors
- The start date of the study is delayed because of logistical problems, and is changed into Mai 15th 2013.